Author’s response to reviews

Title: Metachronous solitary splenic metastasis arising from early gastric cancer: a case report and literature review

Authors:

Tsutomu Namikawa (tsutomun@kochi-u.ac.jp)

Yasuhiro Kawanishi (jm-ykawanishi@kochi-u.ac.jp)

Kazune Fujisawa (kfujisawa@kochi-u.ac.jp)

Eri Munekage (eri-munekage@kochi-u.ac.jp)

Masaya Munekage (m-munekage@kochi-u.ac.jp)

Takahito Sugase (tsugase@gesurg.med.osaka-u.ac.jp)

Hiromichi Maeda (hmaeda@kochi-u.ac.jp)

Hiroyuki Kitagawa (kitagawah@kochi-u.ac.jp)

Makoto Hiroi (hiroim@kochi-u.ac.jp)

Michiya Kobayashi (kobayasm@kochi-u.ac.jp)

Kazuhiro Hanazaki (hanazaki@kochi-u.ac.jp)

Version: 2 Date: 01 Jul 2017

Author’s response to reviews:

July 2, 2017

Dear Editor,

Please find enclosed a copy of our revised manuscript in Word format.

Title: Metachronous solitary splenic metastasis arising from early gastric cancer: a case report and literature review
Thank you for your kind letter of June 7, 2017 regarding our manuscript. The comments were extremely helpful and we have revised our paper accordingly.

A list of the point-by-point responses to reviewer comments are summarized on a separate sheet.

We hope that the revised manuscript is now suitable for publication in BMC Surgery.

Yours sincerely,

Tsutomu Namikawa M.D., Ph.D.
Department of Surgery, Kochi Medical School
Oko-cho, Nankoku, Kochi 783-8505, Japan
(TEL) +81-88-880-2370
(FAX) +81-88-880-2371
E-mail: tsutomun@kochi-u.ac.jp
Response to Reviewers

Reviewer

Thank you for your comments and suggestions.

1. We revised the manuscript in order to lower the similarity score.

Red font on page 5, lines 2 to 10, page 9, lines 6 to 8, lines 11 to 14, page 10, lines 2 to 5